Author's response to reviews

Title: Behavioural interventions for weight management in pregnancy: A systematic review of quantitative and qualitative data.

Authors:

Fiona Campbell (f.campbell@sheffield.ac.uk)
Maxine Johnson (m.johnson@sheffield.ac.uk)
Josie Messina (j.messina@shef.ac.uk)
Louise Guillaume (l.guillaume@sheffield.ac.uk)
Elizabeth Goyder (e.goyder@sheffield.ac.uk)

Version: 6 Date: 25 May 2011

Author's response to reviews:

A number of corrections of small typographical errors have been made. The labels figure 1 and figure 2 have also been changed. I have also made a change to figure 1 which I have reloaded.